Author’s response to reviews

Title: A population-based screening study for cardiovascular diseases and diabetes in Danish postmenopausal women: Acceptability and prevalence

Authors:

Marie Dahl (marie.dahl@rm.dk)
Lars Frost (larfrost@rm.dk)
Rikke Søgaard (rs@ph.au.dk)
Ib Klausen (Ib.C.Klausen@viborg.RM.dk)
Vibeke Lorentzen (vibeke.lorentzen@Viborg.RM.dk)
Jes Lindholt (jes.sanddal.lindholt@rsyd.dk)

Version: 2 Date: 23 Jan 2018

Author’s response to reviews:

To BMC Cardiovascular Disorders,

Thank you very much for the comments on our manuscript.

We have addressed the amendments below:

Comment 1: Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

Revised: "The study was approved by the Regional Scientific Ethical Committee, a part of the Danish National Committee on Health Research Ethics (record number M-20100116) and the Danish Regional Data Protection Agency (record number 1-16-02-221-16)" (page 17).

Comment 1: Please include a keywords section under the abstract. This should include Three to ten keywords representing the main content of the article.
Revised "Keywords: observational study, screening, peripheral arterial diseases, abdominal aortic aneurysm, diabetes mellitus type 2, hypertension, atrial fibrillation, carotid plaque" (page 4).